Author's response to reviews

Title: Acute scrotum as a complication of Thiersch operation for rectal prolapse in a child

Authors:

Mohammad M Saleem (mohomari@hotmail.com)
Hashem Al-Momani (hashemmomani@yahoo.com)

Version: 6 Date: 26 December 2006

Author's response to reviews: see over
Covering letter

Dear Sir

I have responded to the corrections demanded by the journal

1. The abstract has been arranged according to your suggestion and in agreement with the journal format
2. The competing interest statement has been removed from the title page and added after the conclusion section as you suggested
3. The references numbers have been put in brackets
4. The introduction has been changed to background
5. Case report has been changed to case presentation

I hope this meets the corrections you suggested
Thank you for guidance

Sincerely yours
M M Saleem MD